

Evolution of plastic transmission strategies in avian malaria

Stéphane Cornet, Antoine Nicot, Ana R Rivero, Sylvain Gandon

▶ To cite this version:

Stéphane Cornet, Antoine Nicot, Ana R
 Rivero, Sylvain Gandon. Evolution of plastic transmission strategies in avian malaria. PLoS Pathogens, 2014, 10 (9), pp.e1004308.
 10.1371/journal.ppat.1004308. hal-02361650

HAL Id: hal-02361650 https://hal.science/hal-02361650

Submitted on 1 Jun 2021

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License

roseMarl

Evolution of Plastic Transmission Strategies in Avian Malaria

Stéphane Cornet^{1,2}, Antoine Nicot^{1,2}, Ana Rivero², Sylvain Gandon^{1*}

1 Centre d'Ecologie Fonctionnelle et Evolutive (CEFE), UMR CNRS 5175 - Université de Montpellier - Université Paul-Valéry Montpellier - EPHE, Montpellier, France, 2 Maladies Infectieuses et Vecteurs: Ecologie, Génétique, Evolution et Contrôle (MIVEGEC), UMR CNRS 5290-IRD 224-UM1-UM2, Montpellier, France

Abstract

Malaria parasites have been shown to adjust their life history traits to changing environmental conditions. Parasite relapses and recrudescences—marked increases in blood parasite numbers following a period when the parasite was either absent or present at very low levels in the blood, respectively—are expected to be part of such adaptive plastic strategies. Here, we first present a theoretical model that analyses the evolution of transmission strategies in fluctuating seasonal environments and we show that relapses may be adaptive if they are concomitant with the presence of mosquitoes in the vicinity of the host. We then experimentally test the hypothesis that *Plasmodium* parasites can respond to the presence of vectors. For this purpose, we repeatedly exposed birds infected by the avian malaria parasite *Plasmodium relictum* to the bites of uninfected females of its natural vector, the mosquito *Culex pipiens*, at three different stages of the infection: acute (~34 days post infection), early chronic (~122 dpi) and late chronic (~291 dpi). We show that: (i) mosquito-exposed birds have significantly higher blood parasitaemia than control unexposed birds during the chronic stages of the infection and that (ii) this translates into significantly higher infection prevalence in the mosquito. Our results demonstrate the ability of *Plasmodium relictum* to maximize their transmission by adopting plastic life history strategies in response to the availability of insect vectors.

Citation: Cornet S, Nicot A, Rivero A, Gandon S (2014) Evolution of Plastic Transmission Strategies in Avian Malaria. PLoS Pathog 10(9): e1004308. doi:10.1371/ journal.ppat.1004308

Editor: Kenneth D. Vernick, Institut Pasteur, France

Received January 27, 2014; Accepted July 2, 2014; Published September 11, 2014

Copyright: © 2014 Cornet et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: The work was funded by the CNRS and the ERC Starting Grant 243054 EVOLEPID to SG. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* Email: sylvain.gandon@cefe.cnrs.fr

Introduction

All organisms experience some level of temporal variation in the quality of their environment. In response to these variations, many species have evolved specific strategies that allow them to limit or shut down growth and development until the conditions improve [1]. The best reported examples are dormancy in plants and diapause in insects, but similar strategies have also evolved in microbes. Bacteria can survive adverse conditions (e.g. desiccation, antibiotics) by entering a state of reduced metabolic activity called persistence [2,3]. Several viruses (e.g. lambdoid phages, herpesviruses) have evolved the ability to integrate their host genome and enter a latent phase during which within-host replication is shut down, the infection is asymptomatic and transmission is very limited [4,5]. Hence, the evolution of latent life cycle in pathogens may be viewed as an adaptation to temporal variations of the availability of susceptible hosts.

For vector-borne pathogens the abundance of vectors is a key parameter determining the quality of their environment. Vector density may vary in space due to intrinsic heterogeneities of their habitat (e.g. temperature, hygrometry). In malaria, for instance, spatial variation in mosquito abundance has a direct impact on the geographic distribution of prevalence [6–8]. Vector abundance may also vary widely through time [9]. Although inter-tropical regions are characterized by a relatively constant density of vectors, regions from higher latitudes experience a broad range of climatic seasonality, and very far from the equator mosquitoes are present for only a fraction of the year [10–12]. From the parasite's perspective, such temporal variation in vector density is analogous to the temporal variations in habitat quality experienced by other organisms. How have malaria parasites adapted to these temporal fluctuations in vector density?

Malaria is caused by *Plasmodium* spp., a prevalent vector-borne pathogen which is found infecting many vertebrate hosts, including humans, reptiles and birds. Plasmodium infections within the vertebrate host are characterized by drastic temporal changes in blood parasitaemia. After an initial acute phase, generally characterized by a very high number of parasites in the blood, the infection usually reaches a chronic phase where the parasitaemia stabilizes at low levels. During the chronic phase, however, blood parasites may go through short, intense, bouts of asexual replication during which parasitaemia increases temporarily. Little is known about the causes of such recrudescences but one potential trigger may be a weakening of the host's immunity [13]. In some, but not all, *Plasmodium* species the infection may entirely disappear from the blood stream, hiding in other host cells in the form of (dormant) exoerythrocytic stages. After a period of latency that can last months or even years, parasites may reappear in the blood stream. These *relapses* are due to the differentiation of dormant parasite stages into new erythrocytic stages. The dormant stages of *Plasmodium* were first described in birds [14,15] and, later, in humans [16,17] and reptiles [18,19]. Relapses and recrudescences have been puzzling researchers ever since the first clinical symptoms were described in P. vivax-infected humans in the late 19^{th} century [20,21]. Why do some malaria species (e.g. P.

Author Summary

Seasonal fluctuations in the environment affect dramatically the abundance of insect species. These fluctuations have important consequences for the transmission of vector-borne diseases. Here we contend that malaria parasites may have evolved plastic transmission strategies as an adaptation to the fluctuations in mosquito densities. First, our theoretical analysis identifies the conditions for the evolution of such plastic transmission strategies. Second, we show that in avian malaria *Plasmodium* parasites have the ability to increase transmission after being bitten by uninfected *Culex* mosquitoes. This demonstrates the ability of *Plasmodium* parasites to adopt plastic transmission strategies and challenges our understanding of malaria epidemiology.

falciparum) completely lack the ability to produce dormant stages in the vertebrate host? What are the ultimate causes of the production of recrudescences and relapses? Is this diversity of life cycles due to the temporal variation in vector density?

The ability to produce recrudescences and relapses may be a genetically fixed parasite strategy that has evolved as a way to match the dynamics of vector populations. Populations exposed to different fluctuations of vector density may thus evolve different strategies. In human malaria, the relapsing periodicity of different lineages of *P. vivax* supports this prediction [12,22]. Indeed, lineages exhibiting frequent relapses have been sampled in Asia where the vector is present throughout the year. In contrast, long latency has been observed in lineages sampled in temperate zones where the mosquito vector is only present for a few months. In avian malaria, similarly, the differences in the within-host dynamics of *Leucocytozoon spp.* and *Haemoproteus mansoni* may have evolved to match the temporal fluctuations of their respective vector species (simuliid flies and *Culicoides*, respectively) [23].

Another explanation for these patterns may involve adaptive phenotypic plasticity. Phenotypic plasticity is the ability for a single genotype to exhibit different phenotypes in different environments [24,25]. This contrasts with the above hypothesis (fixed strategy) where different relapsing strategies are associated with different genotypes. The ability to adopt a plastic exploitation strategy requires the ability to detect a change of the environment (i.e. cues) and the acquisition of such a sensing mechanism may be associated with direct fitness costs [24,25]. In spite of these costs, phenotypic plasticity is often viewed as an adaptation to variable environments [24,25]. Many pathogens have indeed evolved an unparalleled level of phenotypic plasticity in their life history traits to cope with the temporal variability of their habitat [26-28]. In Plasmodium, plasticity has been shown to be a response to various stressful conditions such as drug treatment and the presence of competitors [29,30]. Some experimental evidence suggests that relapses may also be a plastic trait. P. vivax relapses are often observed in the spring and summer months irrespective of when the patients got the original infection [31], which suggests that the parasite may react to a change in the physiological state of the host or the environment. Relapses have also been observed in avian malaria, which has triggered several experimental studies to pinpoint the underlying environmental cues [32]. Some authors have proposed that spring relapses may result from increasing photoperiod and/or stressinduced hormonal changes [33-36]. Parasites may indirectly benefit from using hormonal and photoperiod cues because they often coincide with (or even anticipate) the appearance of vectors in temperate populations. Such indirect cues are, however, imperfect because vector abundance may be influenced by other, non-seasonal, factors. A more efficient strategy would be to react to direct cues such as mosquito bites which unambiguously indicate the presence of vectors [10,31,37]. Although there is some correlational evidence supporting this hypothesis, largely coming from longitudinal cohort studies [10,37,38], direct experimental evidence for this hypothesis is scarce and somewhat contradictory. In rodent malaria *P.chabaudi*, mice exposed to probing by Anopheles stephensi mosquitoes had higher and earlier parasite growth and gametocytaemia than control unexposed mice [39]. In contrast, however, Shutler et al. [40] found no evidence of facultative alteration in the timing or in the level of *P*. chabaudi or P. vinckei parasitaemia and gametocytogenesis as a consequence of mosquito probing. Rodent malaria is a laboratory model and, as such, may, not be the best system to test this hypothesis because An. stephensi is not the natural vector of rodent malaria [41]. In addition the parasites have been originally sampled from the tropical lowlands of the Congo Basin [42] an area where malaria transmission is high throughout the year [43] and thus the selective pressure for the evolution of plasticity in response to vector availability is expected to be weak. Finally, both rodent malaria experiments [39,40] were carried out during the initial (acute) phase of the infection, i.e. when parasitaemia is already high (so no need to increase it further) and the infection recent (so the mosquitoes are probably still around). We contend that it is mainly in old (chronic state) infections that the parasite may accrue the greatest benefits from a plastic response to the bites of its vector. Finally, both of these studies used gametocyte density (the blood stages of *Plasmodium* that are transmissible to the vector) as a proxy for transmission but neither followed transmission all the way to the mosquito stage.

Here, we first present a theoretical model that studies the evolution of parasite transmission in a variable environment. This model explores the effects of the seasonality of mosquito dynamics on the evolution of virulence and transmission strategies. In particular it clarifies the selective pressures acting on the evolution of temporally variable transmission strategies and identifies the conditions driving the evolution of costly plastic transmission strategies triggered by the exposure to mosquito bites. Then, we carry out an experiment to test the following two hypotheses: (1) *Plasmodium* parasites plastically react to the biting of uninfected vectors by enhancing their within-host replication, and (2) this effect yields higher rates of transmission to the mosquito vector. For this purpose, we studied the interaction between *Plasmodium relictum* (the aetiological agent of the most prevalent form of avian malaria which is commonly found infecting Passeriform birds in Europe) and its natural vector, the mosquito Culex pipiens. P. relictum is a very convenient malaria parasite to address this issue because it is known to have a long chronic phase marked by sudden events of recrudescences and relapses [44]. Strictly speaking, relapses originate from the division and differentiation of dormant stages (called phanerozoites) that infect the endothelial cells of different organs such as the spleen and liver, while recrudescences originate from an increased replication of the blood stages [44]. In practice, however, it is very difficult to distinguish between recrudescences and true relapses and in the following we will use the term relapse to encompass both cases. We investigate whether bites of uninfected Cx. pipiens mosquitoes trigger parasite relapses in the blood of domestic canaries (Serinus canaria) chronically infected by P. relictum (lineage SGS1), as well as the concomitant effects on transmission in terms of mosquito infectivity (see Box 1 and Fig. 1).

Results

Theory: Evolution of plastic transmission strategies

To model the evolution of plastic transmission strategies we first need to model the epidemiological dynamics of malaria. For the

Box 1. Experimental design.

Twenty birds experimentally inoculated with avian malaria parasite *Plasmodium relictum* were followed for over 300 days post-infection (dpi) to monitor the variation in blood parasitaemia. Birds were either exposed or unexposed (control) to mosquito bites. Exposure to mosquito bites took place in 3 consecutive "exposure sessions" (grey area in figure 1). Each session consisted of 3 "exposure days" separated by 3-day intervals: days 34, 37 and 40 dpi for the first session, days 122, 125 and 128 dpi for the second session, and days 291, 294 and 297 dpi for the third session. In exposure days, each bird in the "exposed" treatment was placed in a cage with a batch of 50 uninfected female mosquitoes for 2 hours; "control" birds were placed in identical conditions but without mosquitoes. Two different response variables were subsequently obtained:

- (i) To estimate changes in blood parasitaemia due to mosquito exposure, blood samples were taken from all ("exposed" and "control") birds in the morning preceding the exposure as well as 3–4 days and 7–8 days afterwards (days 44 and 48 dpi, days 131 and 135 dpi and days 300 and 304 dpi for exposure sessions one, two and three, respectively). These are indicated by red arrows in Fig. 1. Bird parasitaemia was measured by qPCR on blood samples (see materials and methods for details). Regular monitoring of parasitaemia took place at several other time points before and between each of the exposure sessions (indicated by blue arrows in Fig. 1).
- (ii) To estimate the effect of exposure on the prevalence and intensity of mosquito infections, after each exposure blood-fed mosquitoes from all the batches were maintained under standard laboratory conditions for 7 days. Fifteen haphazardly chosen mosquitoes were dissected to check for the presence (prevalence) and number (intensity) of oocysts in the midgut (see materials and methods).

sake of simplicity the vertebrate host population is assumed to be constant and equal to N = S(t)+I(t), where S(t) and I(t) are the densities of uninfected and infected hosts, respectively. In contrast, the density of the vector population may change through time. This may be particularly relevant in temperate environments where the mosquitoes do not reproduce in winter. In other words, the influx $\theta(t)$ of uninfected mosquitoes is assumed to change throughout the year (i.e. $\theta(t)$ fluctuates with a period T). As a consequence, the densities V(t) and $V_I(t)$ of uninfected and infected vectors also fluctuate through time. The following set of differential equations describes the temporal dynamics of the different types of hosts (the dot notation indicates differential over time):

$$S(t) = N - I(t)$$

$$\dot{I}(t) = S(t)V_I(t)\beta_2 - (d + \alpha)I(t)$$

$$\dot{V}(t) = \theta(t) - I(t)V(t)\beta_1 - mV(t)$$

$$V_I(t) = I(t)V(t)\beta_1 - m_IV_I(t)$$
(1)

Where *d* is the natural mortality rate of the vertebrate host and α is the virulence of malaria (the extra mortality induced by the infection); *m* and *m_I* are the mortality rates of uninfected and infected vectors, respectively; β_I is the transmission rate from the vertebrate host to the vector; β_2 is the transmission rate from the vector to the vertebrate host. Figure 2 presents a typical epidemiological dynamics in a seasonal environment.

What are the consequences of the temporal variation in the availability of vectors on the evolution of malaria? More specifically, what is the effect of the shape of the function $\theta(t)$ on the evolution of the parasite? To study this question one can consider the fate of a mutant malaria strategy M that would alter its life history strategy in the vertebrate host. The replication in the vertebrate host is assumed to be governed by two traits of the parasite. The first trait, ε_F , governs the allocation to a *fixed* exploitation strategy that yields a within-host growth rate that does not vary with time. In contrast, the second trait, ε_P , governs allocation to a *plastic* exploitation strategy that may vary with time. More specifically we consider that when the parasite adopts this plastic trait within-host growth rate depends on the density of vectors in the population. In other words this plastic trait allows within-host growth rate to be conditional on the presence of vectors. Because within-host growth rate is assumed to affect virulence in the vertebrate host this yields:

$$\alpha(t) = \varepsilon_F + \varepsilon_P(V(t) + V_I(t)) \tag{2}$$

As in classical models of virulence evolution [45,46] more replication is costly because it may induce the death of the vertebrate host but it allows the parasite to transmit more efficiently. The parameters *a* and *b* govern the specific shape of the virulence-transmission trade-off (see equation (3) below). In addition we assume that the adoption of a plastic exploitation strategy requires the ability to acquire information regarding the availability of the vectors. The parameter *c*, therefore measures the fitness cost associated with a higher investment in the mechanisms allowing such plasticity. Only the transmission rate, β_1 , from the vertebrate host to the vector is assumed to be affected by the parasite exploitation strategy (i.e. β_2 is assumed to be constant) which yields:

$$\beta_1(t) = a(\alpha(t))^b - c\varepsilon_P \tag{3}$$

Note that in this model virulence and transmission vary in time only if the parasite allocates some resources in the development of a plastic trait (i.e. $\varepsilon_P > 0$).

Integrating the change in frequency of the mutant parasite genotype M over one period of the fluctuation allows deriving a condition for the invasion of the mutant (see Text S1):

$$s_{M} = \underbrace{\frac{\beta_{2}}{mm_{I}}\widetilde{S\theta}\Delta\beta_{1}}_{Benefit of transmission} \underbrace{-\Delta\alpha}_{Cost of} + \underbrace{\frac{\beta_{2}}{mm_{I}}\Delta COV}_{Benefit of plasticity} > 0 \quad (4)$$

where
$$\widetilde{S\theta} = \frac{1}{T} \int_0^T S(t)\theta(t)dt$$
, $\Delta\beta_1 = \frac{1}{T} \int_0^T (\beta_{1M}(t) - \beta_1(t))dt$,
 $\Delta\alpha = \frac{1}{T} \int_0^T (\alpha_M(t) - \alpha(t))dt$ and $\Delta COV = COV_t(S\theta, \beta_{1M}) - COV_t(S\theta, \beta_1)$.

The first term in the above equation for s_M is the classical benefit associated with higher investment in transmission. If the



Figure 1. (A) Overview of the experiment with the 3 exposure sessions (grey areas). Arrows indicate the times at which blood samples were taken from the birds: red arrows for the 5 samples taken in and around the time of the mosquito exposure, blue arrows for the regular monitoring of parasitaemia before and between exposures. One unit = 10 days. (B) Zoom on an exposure session: each rectangle represents a day. Black rectangles: blood sample in the morning, mosquito exposure in the evening. Grey rectangles: blood sample in the morning, no mosquito exposure. White rectangles: no mosquito exposure, no blood sampling. Red arrows and figures underneath indicate dates where blood sampling took place in each of the 3 sessions. The mosquito drawing indicates a mosquito exposure.

mutant invests more than the resident in transmission (i.e. $\Delta\beta_1 > 0$) the fitness increase depends on $\widetilde{S\theta}$, which measures the availability of both uninfected hosts and vectors over the period of the fluctuation of the environment. The second term in s_M is the classical cost of virulence if the mutant exploits the host more aggressively than the resident (i.e. $\Delta \alpha > 0$). The final term in s_M measures the potential benefit associated with plastic transmission strategies. This term depends on the covariance between the availability of uninfected vectors, the availability of uninfected vertebrate hosts and the investment in transmission from vertebrate to mosquito hosts. The mutant may gain a fitness advantage if its conditional transmission rate can better track the fluctuations of the density of uninfected hosts. In other words this final term indicates that in a fluctuating environment it is adaptive to invest on transmission whenever uninfected hosts and mosquitoes reach high densities simultaneously. We can use this analysis to look at different evolutionary scenarios.

Without plasticity. For the sake of simplicity let us focus first on the influence of seasonality on the evolution of the fixed exploitation strategy in the absence of plasticity. In this case $\varepsilon_P = 0$ and neither virulence nor transmission vary in time. Consequently the final term in s_M drops and evolution is driven by the balance between the benefit of transmission and the cost of virulence. In our model the cost of virulence is not affected by the epidemiological dynamics (infected hosts die because of the infection irrespective of the presence of the vectors). In contrast, the benefit of transmission is weighted by the quantity $\tilde{S}\theta$ which measures the overall opportunity of contacts between uninfected hosts and vectors. Fluctuations in θ over time may not necessarily affect the quantity $\widetilde{S\theta}$ and in these situations seasonality has no impact on virulence evolution (Text S1). Yet, contrasting situations with or without wintering season dramatically affects the average influx of vectors and consequently the quantity $\widetilde{S\theta}$. To study the effect of seasonality we model the fluctuations of the influx of uninfected mosquitoes as a periodic square wave (Fig. 2):

$$\theta(t) = A H\left(Sin\left(\frac{2\pi}{T}t\right) - Cos(\pi(1-\tau))\right)$$
(5)

Where A is the influx of uninfected mosquitoes when the environment is favorable for mosquito reproduction, T is the period of the fluctuations, τ is the fraction of time unsuitable for mosquito reproduction and H(x) is the discontinuous unit step function taking the value 0 (when x < 0) or 1 (when x > 0). In Figure 3 we show that when seasonality increases the quantity $\widetilde{S\theta}$ drops, the benefit of transmission is reduced and the parasite evolves toward lower virulence and lower transmission rates. In other words our model predicts that, away from the tropics, malaria population will experience more seasonal environments and the investment in the fixed level of host exploitation drops to avoid the cost of virulence when no vectors are around. This should yield lower levels of transmission and virulence in higher latitudes.

With plasticity. Our model can also be used to understand the conditions leading to the evolution of plastic transmission strategies. In this case we allow both fixed and plastic exploitation strategies (i.e. ε_F and ε_P) to evolve freely. Figure 4 shows that when



Figure 2. Epidemiological dynamics of a vector-borne disease in a seasonal environment. We consider here that the influx of mosquitoes, $\theta(t)$, is a periodic square wave (see equation (5) in the main text). The parameter *T* measures the duration of the period and the parameter τ measures seasonality: the fraction of time where the environment is not suitable for vector reproduction. The epidemiologic dynamics converges to a periodic equilibrium characterised by fluctuations of the uninfected and infected vector the densities: V(t) and $V_i(t)$ (black and red dashed lines, respectively). We also plot the dynamics of the density of infected hosts: I(t)/10 (red line). Parameter values: $\varepsilon_F = 1$, $\varepsilon_P = 0$, $\beta_2 = 0.5$, $\tau = 0.5$. See default values in the Text S1 for other parameters.

doi:10.1371/journal.ppat.1004308.g002



Figure 3. Evolution of allocation to fixed pathogen strategy, ε_{Fr} as a function of seasonality, τ . A higher investment in ε_F indicates that the pathogen invests more in transmission (and virulence). Parameter values: $\varepsilon_P = 0$. See default values in the Text S1 for the other parameters. doi:10.1371/journal.ppat.1004308.g003

the vectors are present throughout the year the pathogen does not invest in a costly plastic strategy. In this case, the parasite population evolves toward the fixed level of host exploitation studied in the previous scenario and resulting from the balance between the benefit and the cost of parasite virulence. When the environment becomes more seasonal (i.e. $\tau > 0$) the allocation to a fixed exploitation strategy drops and the pathogen invests in the costly plastic strategy (Fig. 4). The plastic strategy allows allocating resources to host exploitation only when vectors are around. Hence this relatively simple model confirms that a costly plastic transmission strategy that depends on the availability of vectors in the environment can outcompete a constant exploitation strategy. Note that for intermediate levels of seasonality the evolutionarily stable exploitation strategy is a mixture between a constant and a plastic strategy.

Experiment with avian malaria

The experimental design is presented in Box 1. In brief, we followed 20 experimentally infected birds over 300 days post infection and monitored within-host parasitaemia and transmission to vectors. Birds were assigned to two treatments: "exposed" or "control" (unexposed) to uninfected mosquito bites during 3 sessions (starting 34, 122 and 291 days post infection, see Fig. 1A). During each session the exposed birds were bitten by a batch of 50 female mosquitoes every 3 days (see Fig. 1B).

Parasitaemia. The parasitaemia initially followed a bellshape function typical of acute *Plasmodium* infections: peaking at day 14 pi and decreasing thereafter (Fig. 5). The infection subsequently entered a long-lasting chronic state, which was characterized by a low (but detectable) blood parasitaemia over several months (Fig. 5).

The effect of mosquito exposure on blood parasitaemia was analysed separately for each of the 3 exposure sessions. In the first (34–48 dpi) session, parasitaemia was still decreasing after the initial (acute) phase (*time* effect: $\chi^2_1 = 11.14$, P = 0.0008) but this decrease was independent of whether the birds had been exposed

to mosquitoes or not (exposure effect: $\chi^2_1 = 0.007$, P = 0.9345) (Fig. 6A). In the second (122–135 dpi) session, however, bird parasitaemia differed between the exposed and the unexposed (control) birds. Whereas in the control birds the total number of parasites remained roughly constant with time, parasitaemia in the exposed group increased significantly with time (exposure*time: $\chi^2_1 = 9.18$, P = 0.0024, Fig. 6B). In the third (291–304 dpi) session, parasitaemia showed a similar trend towards a higher parasitaemia in exposed mosquitoes (Fig. 6C), but this trend was not statistically significant (exposure: $\chi^2_1 = 0.40$, P = 0.5270; time: $\chi^2_1 = 18.87$, P = 0.0003). At this time point, however, several birds had died, which reduced the statistical replication and limited the statistical power of the test (n = 6 birds alive on day 291, n = 3 and n = 4 on day 304 for exposed and control birds, respectively) (see Table S1 for group sample sizes).

It is worth noting that the effect of exposure to mosquito bites seems to be short-lived. Indeed, one month after the second exposure session (165 dpi) we did not detect any difference in parasitaemia between the exposed and the control birds ($\chi^2_1 = 0.03$, P = 0.5932, see Fig. 5).

Mosquito infection. In the first exposure session, infection prevalence (proportion of mosquitoes containing at least 1 oocyst) was extremely high among the first batch of mosquitoes (Table S2) but decreased in subsequent batches (contrast 34+37*vs*. 40 dpi, $\chi^2_1 = 51.71$, P < 0.0001; Fig. 7A). This effect is linked to the decrease in overall parasitaemia ($\chi^2_1 = 48.45$, P < 0.0001) and is likely due to the fact that the first session occurred at the end of the acute phase and before the start of the chronic phase (see Fig. 5). In contrast, in both the second and third exposure sessions infection prevalence increased significantly in successive mosquito batches (second exposure session: contrast 122 *vs*. 125+128 dpi, $\chi^2_1 = 66.34$, P < 0.0001; third exposure session: $\chi^2_2 = 25.99$, P < 0.0001, here all time points differed from each other, Fig. 7B and B).



The analysis of oocyst burden only included mosquitoes having one or more oocysts in the midgut. Oocyst burden showed a

Figure 4. Joint evolution of (A) the plastic pathogen strategy, ε_P and of (B) the fixed pathogen strategy, ε_F for different values of seasonality, τ , and for different costs of plasticity, c. The color shading indicates the value of the pathogen strategies and the warmer color indicates higher values. A higher investment in ε_P indicates that the pathogen invests more into the mechanisms that allow it to react to the presence of mosquitoes. A higher investment in ε_F indicates that the pathogen invests more into transmission (and virulence). For both strategies the lower value (blue) is 0. The maximal value (red) of ε_P is 4 and the maximal value (red) of ε_F is 1.1. See default values in the Text S1 for the other parameters. doi:10.1371/journal.ppat.1004308.g004



Figure 5. Dynamics of blood parasitaemia (Log(RQ+1), mean ± s.e.) of *Plasmodium relictum* (lineage SGS1) in birds that were either unexposed (open circles, dashed line) or exposed to mosquito bites (filled circles, solid line). Mosquito exposure (refer to Materials & Methods for details) took place in three consecutive sessions (grey areas): 34–48 dpi, 122–135 dpi and 291–304 dpi. doi:10.1371/journal.ppat.1004308.g005

consistent pattern across the three exposure sessions: the number of oocysts increased significantly between the first and second mosquito batches but decreased thereafter (*batch time* effect: $\chi^2_2 = 1147$, P < 0.0001, $\chi^2_2 = 546.17$, P < 0.0001 and $\chi^2_2 = 389.84$, P < 0.0001 for the first, second and third exposure sessions respectively; all contrast analyses were significant; Fig. 8).

To verify that the control birds were still infective to mosquitoes, they were exposed to mosquitoes on day 307 pi. Only three birds survived to this point and all three were infective to mosquitoes (Table S2). The infection rate of mosquitoes biting the control birds was compared to the infection rate of the last batch of mosquitoes biting the exposed birds (297 dpi). As expected from the above



Figure 6. Details of the dynamics of blood parasitaemia (Log(RQ+1), mean \pm s.e.) for the 3 exposure sessions (see Fig. 4): (A) session 1 (34–48 dpi), (B) session 2 (122–135 dpi) and (C) session 3 (291–304 dpi). Unexposed (open circles, dashed line) or exposed to mosquito bites (filled circles, solid line). doi:10.1371/journal.ppat.1004308.g006



Figure 7. Boxplot of the proportion of infected mosquitoes among 15 haphazardly chosen blood fed individuals on each bird (harbouring at least 1 oocyst in the midgut) for the 3 exposure sessions (see Fig. 4): (A) session 1 (34–40 dpi), (B) session 2 (122–128 dpi) and (C) session 3 (291–297 dpi). The figure shows the median proportion of infected mosquitoes (horizontal black bars). The white boxes below and above the median indicate the first and third quartiles respectively. Dashed lines delimit 1.5 times the inter-quartile range on both side of the box, above which individual counts are considered outliers and marked as dots. doi:10.1371/journal.ppat.1004308.g007

results, the infection rate of mosquitoes biting a bird for the first time was significantly lower than the infection rate of mosquitoes biting a bird which has been recently bitten by two batches of mosquitoes in previous days (infection rate of mosquitoes biting the control birds: 0.36 ± 0.12 , exposed birds 0.58 ± 0.10 ; $\chi^2_1 = 5.76$, P = 0.0164).

Discussion

Plasticity has evolved as an adaptation to the variability of the environment in many organisms [25,47], including pathogens [27,28,48,49]. Here we contend that the evolution of fixed or plastic dormancy strategies in *Plasmodium* may be an adaptation to the seasonal fluctuations of vector densities. We explore this hypothesis with a theoretical model and test experimentally some of our predictions in avian malaria.

Theory

How do malaria parasites adapt to the density fluctuations of their insect vectors? To answer this question we started by studying the evolution of transmission strategies using a classical epidemiological model for a vector-borne pathogen. This theoretical approach helps clarify the multiple effects of temporal fluctuations of vector populations. We first considered the evolution of a fixed allocation to virulence and transmission. Our analysis shows that the effect of the temporal variation is driven by its effect on the average density of susceptible hosts and vectors over one period of the fluctuation. In particular we show that in more seasonal environments (e.g. higher latitudes), where the vectors can pullulate only for a few months, lower levels of virulence and transmission should be selected. This is because, in our model, seasonality reduces the average number of vectors. In the absence of the vector, investing in transmission becomes maladaptive because within-host reproduction is associated with higher virulence and host death. This result is very similar to the effect of periodic host absence on the evolution of phytopathogens when there is a trade-off between pathogen transmission and pathogen survival [50]. In addition, our predictions agree with recent models studying the effect of seasonality on virulence evolution [51], in that if the fluctuations of vector density do not affect the mean



Figure 8. Boxplot of the number of oocysts per midgut among 15 haphazardly chosen blood fed individuals on each bird (only includes mosquitoes harbouring ≥1 oocysts) for the 3 exposure sessions (see Fig. 4): (A) session 1 (34–40 dpi), (B) session 2 (122–128 dpi) and (C) session 3 (291–297 dpi). doi:10.1371/journal.ppat.1004308.g008

density of susceptible vectors over time, we expect no evolutionary consequences. Interestingly, our prediction on the effect of seasonality (Fig. 3) is consistent with the geographical distribution of relapsing strategies in *P. vivax* [22]. *P. vivax* genotypes sampled near the equator (where seasonality is minimal) invest in higher transmission strategies (higher rates of relapse) than *P. vivax* genotypes sampled in higher latitudes. In other words, in *P. vivax* malaria latitude is a very good predictor of the rate of relapses (Fig. 9).

In a second step of the analysis we allowed plastic transmission strategies to evolve. In particular, we assumed that the malaria pathogens have the ability to sense the density of vectors through exposure to mosquito bites. We derived the condition promoting the evolution of such plastic behaviours when investment in this strategy is associated to a direct fitness cost on transmission. Koelle *et al.* [52] derived a similar result in a model of pathogen adaptation to seasonal fluctuations but without highlighting the force driving adaptive plasticity. Kumo and Sasaki [53] showed that the sensitivity to seasonality in a directly transmitted pathogen is driven by the correlation between the seasonal variation in transmission rate and the density of susceptible hosts. In our model the sensitivity to seasonality is governed by the fluctuation of mosquito density and plasticity. Similarly we show that what selects for plasticity is the covariance between transmission and the availability of hosts (both the vertebrate hosts and the vectors). In other words, plasticity evolves when mosquito bites provide accurate information on the availability of susceptible hosts. Cohen [54] obtained very similar results on the evolution of conditional dormancy strategies in randomly varying environments. The evolution of conditional dormancy depends on the correlation between the cue and the quality of the environment for individuals leaving the dormant state [54] (see also [55,56]). In our model the correlation between the cue (mosquito bites) and the



Figure 9. Effect of latitude on the relapsing rate of *Plasmodium vivax*. The data was obtained from the supplementary Table S1 of Battle *et al.* [22]. Each dot represents a parasite strain originating from different locations. The latitude of origin has a significant effect on the observed time to first relapse ($R^2 = 0.4966$, $F_{1,232} = 228.9$, P < 0.0001). doi:10.1371/journal.ppat.1004308.g009

abundance of susceptible hosts depends on seasonality: as expected, plasticity evolves more readily when mosquitoes are only around for a short period of time.

Experiment

Have malaria parasites evolved the ability to respond plastically to mosquito bites on its vertebrate host? Previous work on acute rodent malaria infections has produced somewhat contrasting results [39,40]. These earlier studies had in common that (i) they used an unnatural mosquito-Plasmodium combination, (ii) they were carried out using parasites collected in a high-transmission tropical environment several decades ago and (iii) were carried out when the infection is already at its highest level within the vertebrate host. Here we use a natural mosquito-Plasmodium combination to test the effect of mosquito bites on parasite transmission during the chronic phase of the infection. We used a P. relictum lineage (SGS1) which had been sampled from wild house sparrows in 2009 in a high latitude habitat (Dijon, France) where the environment is characterized by marked seasonal patterns, including variations in mosquito prevalence [57]. In addition, rather than inferring parasite transmissibility solely from the host's parasitaemia, we also quantified the number of parasites that made it all the way to the gut (oocyst) stages of the biting mosquitoes. Our experiment confirmed our two main predictions. First, P. relictum SGS1 reacts to mosquito bites by increasing its overall parasitaemia in the blood. As expected, this effect was not present during the acute infection (first exposure session) because transmission is always high at this stage, but became apparent during the chronic stage of the infection (second and third exposure sessions). Second, this increased parasitaemia resulted in higher probability of infection to mosquitoes and thus in higher transmission rates. The results were consistent at the chronic stage of the infection (exposure sessions 2 and 3): there was a significant increase in mosquito infection rate after exposure to mosquito bites.

Blood stage malaria infections comprise both asexual (replicating) and sexual (transmissible) stages. However, the molecular tools used to quantify overall parasitaemia in this study did not allow us to distinguish between these two parasite life stages. In other malaria systems the conversion rate between the asexual and the sexual (gametocyte) stages, and the resulting sex ratio of the gametocytes may be highly plastic [30,58], so that overall parasitaemia may not necessarily be a good predictor of gametocyte density and/or transmission. Although nothing is yet known about the conversion rates or sex allocation strategies in P. relictum, in our experiment the increase in parasitaemia was accompanied by a significant increase in the number of infected mosquitoes, suggesting a concomitant increase in gametocyte density in birds exposed to mosquito bites. However, to directly test this hypothesis, we compared the gametocytaemia of exposed and unexposed birds by counting the visible gametocytes in the thick blood smears taken after the exposure (see Text S2). Contrary to expectations, however, we found no clear and consistent evidence that mosquito bites result in higher gametocytaemia. One potential explanation of this lack of consistency is an error in our estimate of gametocytaemia. The application of molecular techniques for the quantification of gametocytes has indeed called into question the use of microscopic methods to quantify Plasmodium gametocytes [59]. In particular, these studies have shown that submicroscopic gametocyte densities are common [60,61] and can readily infect mosquitoes [62]. Unfortunately these molecular tools are currently only available for P. falciparum and P. chabaudi, and no equivalent tools exist to estimate gametocytaemia in P. relictum.

A potential caveat of these results is that all the mosquitoes used in the same exposure session emerged roughly at the same date. As a consequence, females from the second and third mosquito batches were 3 and 6 days older (respectively) than mosquitoes used in the first batch. To control for a potential confounding effect of female age on transmission we therefore carried out another experiment using females of identical age (7 days old) at each exposure session. This experiment was carried out using laboratory (SLAB strain) mosquitoes and although the oocyst infection intensities were overall lower, we obtained qualitatively similar effects as in the main experiment (see Text S3). In addition, earlier studies have found either that age has little effect on mosquito vector competence [63] or that older mosquitoes have a lower prevalence and intensity of infection than their younger counterparts [64]. Hence, the increase in oocyst prevalence and intensity observed in consecutive exposure sessions in our main experiment cannot be explained by differences in the age of the mosquitoes used.

Mechanisms

The proximal mechanism governing this form of plasticity remains to be investigated. How do parasites in the blood or in tissues perceive mosquito bites? A plethora of substances and molecules present in the salivary fluid are injected when mosquitoes probe and feed [65]. The primary role of these molecules is to combat host homeostasis and to regulate inflammation at the biting site to facilitate blood uptake. Vector salivary lysates have been shown to stimulate within-host growth of Leishmania parasites [66] and may also trigger plastic life-history strategies in Plasmodium. In addition, host anaemia, erythropoeisis, and asexual density have all been shown to be associated with the onset of gametocytogenesis in rodent malaria [67-70]. Shutler et al. [40] suggested that blood feeding mosquitoes may cause host anaemia thereby triggering gametocytogenesis in P. chabaudi. Our data, however, do not support this hypothesis, because birds exposed to mosquitoes had similar or even higher haematocrit than control birds (see Text S3). In addition, previous findings obtained using P. gallinaceum have shown that in this avian malaria parasite, parasitaemia and gametocytaemia are not affected by host anaemia [71]. The study of the mechanisms governing plastic transmission strategies in avian malaria is hampered by the lack of available molecular tools to quantify and sex gametocytes (e.g. [72] for rodent malaria). Mechanistic studies can reveal fascinating pathogen strategies. For instance, a recent study on the Cauliflower mosaic virus (CaMV) has shown that when aphids feed on the infected plants the virus reacts instantly (and reversibly) to maximize its transmission. For this purpose it modifies the distribution of a specialised set of proteins which are essential for virus transmission [73]. In the absence of the vector, these proteins, which are toxic for the plant, are neatly packed away inside specialised structures called "transmission bodies". This study not only represents an excellent example of the ability of some vector-borne pathogens to adopt plastic transmission strategies but it also demonstrates the sophisticated molecular and cellular mechanisms that may be involved.

Perspectives

Theory. The theoretical framework we developed could be used to consider other forms of fluctuations of the environment. For instance, at a short time scale, most malaria parasites show a periodicity in replication burst and allocation to parasite transmissible stages. The synchronicity between this periodicity and the fluctuation of vector density is very important for transmission success [74] and Hawking postulated that the periodicity of

malaria may have evolved as a way to maximize the availability of mature gametocytes when mosquitoes feed [75]. Although this hypothesis remains controversial [76,77] our approach could help identify the conditions that may promote the evolution of cell cycle coordination in malaria as a response to daily fluctuations of vector availability.

Our theoretical framework could be readily extended to consider the evolution of other cues triggering higher transmission rates. In the field, co-infections of different *Plasmodium* strains or species are the norm rather than the exception [78,79]. The relapsing strategy of malaria may depend on the presence of other competing pathogens within the host [80]. It would be particularly interesting to investigate the effect of coinfections on the activation of relapses [81,82].

Experimental. Our model predicts that seasonality could have a huge impact on the evolution of pathogens (see Fig. 4). Figure 9 shows data in *P. vivax* supporting our prediction on the evolution of fixed pathogen strategies. Testing our predictions regarding the effect of seasonality on the evolution of plasticity may, however, be more challenging. The well recorded worldwide distribution of avian malaria (MalAvi Database, [83]), however, provides an unparalleled opportunity to test this hypothesis. We predict that tropical lineages should exhibit lower potential for plasticity and lower response to external stimuli (such as mosquito bites) than temperate lineages.

It is important to investigate further the role of the quantity and the quality of the mosquitoes on the reactivation of dormant parasites. Quantity-wise, the threshold number of mosquitoes required for the reactivation to take place should be established. Billingsley *et al.* [39] reported an effect of the amount of mosquitoes biting on gametocytaemia of rodent malaria. Quality-wise, it would be interesting to study if plasticity is specific to particular vector species. Indeed, we may expect *Plasmodium* to respond only to the bites of mosquitoes species that serve as competent vectors of the disease [66]. Alternatively, a general response to the bites of non-vector species may be indicative of an evolutionary constraint (the inability of the parasite to distinguish between vector and non-vector signals), or of a temporal correlation in the abundance of vector and non-vector species in a given area.

Conclusions

We identified the conditions that promote the evolution of plastic transmission strategies in a fluctuating environment. In line with our theoretical predictions, we show that P. relictum has the ability to boost its own transmission during the chronic phase of the vertebrate infection after being exposed to mosquito bites. Whether this ability extends to other Plasmodium species and in particular to human malaria remains to be investigated. In P. vivax the data presented in Figure 9 indicates a strong effect of latitude (i.e. seasonality) on relapses and transmission. The role of plastic transmission strategies on this pattern is unclear but it deserves further investigation. This may help define better control strategies, with more specific recommendations on both spatial and temporal implementations of targeted interventions against malaria hotspots [84]. The study of plastic transmission strategies may also be relevant to many other pathogens that are known to alternate between acute and dormant phases such as varicella zoster virus [85] Herpes Simplex virus [86], Mycobacterium tuberculosis [87] and HIV [88]. Such dormant parasites pose considerable therapeutic challenges and much would be gained from understanding the cues underlying the switch between dormant and acute stages in these pathogens [89-91]. In conclusion, a better understanding of the ecological determinants as well as the evolutionary forces governing parasite relapses is not only of academic interest: it is also urgently needed to improve the efficacy of public health strategies.

Materials and Methods

Ethics statement

Animal experiments were carried out in strict accordance with the National Charter on the Ethics of Animal Experimentation of the French Government, and all efforts were made to minimise suffering. Experiments were approved by the Ethical Committee for Animal Experimentation established by the authors' institution (CNRS) under the auspices of the French Ministry of Education and Research (permit number CEEA- LR-1051).

Malaria parasites and mosquitoes

Plasmodium relictum (lineage SGS1) is a generalist parasite and the most prevalent form of avian malaria in Europe, infecting over 30 birds species in the order Passeriformes [44,83]. Our strain was originally isolated from wild house sparrows caught in the region of Dijon (France) and maintained in the laboratory via passages to naïve canaries either by intraperitoneal injection or through the bite of infected *Culex pipiens* mosquitoes.

Experiments were conducted with wild *Cx. pipiens pipiens* mosquitoes. *Cx. pipiens* is the natural vector of *P. relictum* in the wild [44,92,93]. Thousands of L3 and L4 larvae were collected from a single sewage treatment lagoon in the village of Triadou (20 km north Montpellier, France) using a hand net and reared till adulthood under standard laboratory conditions [94]. We used females 7, 10 and 13 days after emergence that had had no prior access to blood, had been maintained on glucose solution (10%) since their emergence, and had been starved (but provided with water) for 6 h before the experiment.

Experimental design

Experiments were carried out using (1-year old) domestic canaries (Serinus canaria). Prior to the experiments, a small amount (ca. 15-25 µL) of blood was collected from the brachial vein of each of the birds and used for molecular sexing [95], as well as to verify that they were free from any previous haemosporidian infections [96]. Twenty birds were experimentally inoculated on the 3rd July 2010 (day 0, see Box 1 and Fig. 1) by means of an intraperitoneal injection of ca. 50-100 µL of an infected blood pool. The blood pool was constituted of a mixture of blood from 8 infected canaries that had been inoculated with the parasite 12 days previously following standard laboratory procedures [97]. Note that unlike what happens in some *Plasmodium* parasites such as P. vivax, the artificial infection with P. relictum via the inoculation of infected blood containing merozoites does not prevent the formation of exoerythrocytic stages [44,98]. One bird failed to get infected and the remaining infected birds were assigned to two treatments: "exposed" (n = 10) or "unexposed" (control, n = 9) to mosquito bites. This assignment was made by balancing the gender of birds and the magnitude in the peak parasitaemia during the acute phase between the two treatments. This experimental design thus allowed mosquitoes to both probe and blood feed on the birds, and in this respect it contrasts with previous designs where only probing was allowed [39,40].

Exposure to mosquito bites took place in August 2010 (first exposure session), and repeated in November 2010 and April 2011 (second and third exposure sessions). Each of these exposure sessions consisted of 3 "exposure days" separated by 3 day intervals: days 34, 37 and 40 post-infection (dpi) for the first

exposure session, days 122, 125 and 128 dpi for the second exposure session, and days 291, 294 and 297 dpi for the third exposure session (Box 1 and Fig. 1A). In the morning of each exposure day, a small (ca. 15-25 µL) amount of blood was taken from the brachial vein of all ("exposed" and "control") birds to quantify parasitaemia (see below). In the evening, birds allocated to the "exposed" treatment were placed inside a cage (dimensions L40×W30×H30 cm) with a batch of 50 uninfected female mosquitoes for 2 hours (8-10pm). Around 30 females blood fed on the birds during this time (see Table S3) which is close to available estimations in the wild [99]. Tables S2 and S3 provide the full details of the number of replications (number of blood fed mosquitoes, number of mosquitoes dissected) for each exposure session. To minimize host defensive behaviours that may alter the mosquito biting process during the assay, we immobilized birds in a specially designed PVC tube that rendered their legs accessible to the mosquitoes while protecting the rest of the body from the bites [97]. "Control" birds were placed in identical conditions but without mosquitoes. Immediately after each exposure, blood-fed mosquitoes from each cage (n = 10) and time point (3 exposure sessions, 3 days per session) were collected, isolated in a new cage, and maintained under standard laboratory conditions for 7 days. Fifteen haphazardly chosen mosquitoes were dissected to check for the presence (prevalence) and number (intensity) of oocysts in the midgut [94].

In each exposure session, two further blood samples were taken from all experimental birds, 3–4 days and 7–8 days after the last exposure day (days 44 and 48 dpi, days 131 and 135 dpi and days 300 and 304 dpi, for the first to third exposure sessions, Fig. 1B). For each exposure session we therefore obtained 5 different blood samples (red arrows in Fig. 1A and 1B). These blood samples were used to quantify total parasitaemia using previously published qPCR procedures [97] and gametocytaemia by microscopic examination (see Text S2). In addition, blood samples were taken at regular intervals throughout the experiment to monitor parasitaemia before and between the exposure sessions (blue arrows in Fig. 1A).

Statistical analyses

The statistical analyses were run using the R software (v. 2.14.0). Analyses were carried out separately for each exposure session. Variation in parasitaemia (log-transformed (RQ+1)) was analyzed using linear mixed-effect models (*lme* function, *nlme* package) with bird as a random effect to account for the repeated sampling of individual hosts. A generalized linear mixed-effect models GLMM (*glmer* function, *lme4* package, binomial distribution) was carried out to study variation in gametocytaemia (proportion of gametocytes). Bird and time were included as random and fixed factors, respectively.

Variation in the infection prevalence (proportion of individuals harbouring at least 1 oocyst) and the oocystaemia (number of oocysts, only for infected mosquitoes) was analysed using GLMMs (*glmer* function, *lme4* package, with binomial and Poisson distributions, respectively). Bird and time (i.e. time between the 5 different blood samples, see red arrows in Fig. 1b) were included as random and fixed factors, respectively. Here, time was considered as a factorial explanatory variable.

References

- Meyers LA, Bull JJ (2002) Fighting change with change: adaptive variation in an uncertain world. Trends Ecol Evol 17: 551–557.
- Massey RC, Buckling A, Peacock SJ (2001) Phenotypic switching of antibiotic resistance circumvents permanent costs in *Staphylococcus aureus*. Curr Biol 11: 1810–1814.
- Balaban NQ, Merrin J, Chait R, Kowalik L, Leibler S (2004) Bacterial persistence as a phenotypic switch. Science 305: 1622–1625.

When appropriate, *a posteriori* contrasts were carried out by aggregating factor levels that did not significantly differ from each other and by testing the fit of the simplified model [100]. The significance of explanatory variables was established by comparing the change in deviance with and without the term to a χ^2 distribution. Degrees of freedom correspond to the difference in the number of terms in the model.

Supporting Information

Table S1 Sample size for exposed and unexposed groups of birds across the experiment. Time points (days post-infection) refer to sampling times for the monitoring of blood parasitaemia. (DOCX)

Table S2 Table summarizing the number of infected mosquitoes (over the 15 dissected) and oocyst burden (mean \pm s.e.) for the 3 exposure sessions. Unexposed birds were kept as controls during the experiment, the birds that survived (see Table S1) were exposed once to mosquitoes at the end of the experiment (307 dpi). (DOCX)

Table S3 Table summarizing the number of blood fed mosquitoes (around 50 were introduced into each cage) and, in parenthesis, the percentage of blood feeding success for the 3 exposure sessions. Unexposed birds were kept as controls during the experiment, the birds that survived (see Table S1) were exposed once to mosquitoes only at the end of the experiment (307 dpi).

(DOCX)

 $Text\ S1$ Theory - Epidemiology and evolution of inducible transmission strategies.

(DOCX)

Text S2 Experiment - Quantification of gametocytaemia: Temporal variation and relationships with mosquito infection. (DOCX)

Text S3 Experiment - Quantification of the effect of mosquito exposure on parasite transmission: A comparison of the differences in transmission between exposed and control birds. (DOCX)

Acknowledgments

We are grateful to Philippe Perret for his invaluable help with the birds and Troy Day for many useful discussions on adaptive plasticity. We would also like to thank Rick Paul, Sarah Reece and one anonymous referee for their useful comments which greatly helped to improve the manuscript.

Author Contributions

Conceived and designed the experiments: SC AR SG. Performed the experiments: SC. Analyzed the data: SC. Contributed reagents/materials/ analysis tools: AN. Wrote the paper: SC AR SG. Developed and analysed the theoretical model: SG.

- Stewart FM, Levin BR (1984) The population biology of bacterial viruses: why be temperate. Theor Pop Biol 26: 93–117.
- Stumpf MPH, Laidlaw Z, Jansen VAA (2002) Herpes viruses hedge their bets. Proc Natl Acad Sci USA 99: 15234–15237.
- Trape JF, Lefebvre-Zante E, Legros F, Ndiaye G, Bouganali H, et al. (1992) Vector density gradients and the epidemiology of urban malaria in Dakar, Senegal. Am J Trop Med Hyg 47: 181–189.

- 7. Rogers DJ, Randolph SE, Snow RW, Hay SI (2002) Satellite imagery in the study and forecast of malaria. Nature 415: 710-715.
- 8. Mbogo CM, Mwangangi JM, Nzovu J, Gu W, Yan G, et al. (2003) Spatial and temporal heterogeneity of Anopheles mosquitoes and Plasmodium falciparum transmission along the Kenyan coast. Am J Trop Med Hyg 68: 734-742.
- 9. Oesterholt MJAM, Bousema JT, Mwerinde OK, Harris C, Lushino P, et al. (2006) Spatial and temporal variation in malaria transmission in a low endemicity area in northern Tanzania. Malaria J 5: 98.
- 10. Paul REL, Diallo M, Brey PT (2004) Mosquitoes and transmission of malaria parasites - not just vectors. Malaria J 3: e39.
- 11. Ponçon N, Toty C, L'AMBERT G, Le Goff G, Brengues C, et al. (2007) Population dynamics of pest mosquitoes and potential malaria and West Nile virus vectors in relation to climatic factors and human activities in the Camargue, France. Med Vet Entomol 21: 350-357.
- 12. White NJ (2011) Determinants of relapse periodicity in Plasmodium vivax malaria. Malaria I 10: 297.
- 13. McLean SA, Person CD, Phillips RS (1982) Plasmodium chabaudi: relationship between the occurence of recrudescenct parasitaemias in mice and the effective levels of acquired immunity. Exp Parasitol 54: 213-221.
- 14. Huff CG, Bloom W (1935) A malarial parasite infecting all blood and bloodforming cells of birds. J Infect Dis 57: 315-336.
- James SP, Tate P (1937) New knowledge of the life-cycle of malaria parasites. 15. Nature 139: 545.
- 16. Shortt HE, Garnham PCC, Covell G, Shute PG (1948) Pre-erythrocytic stage of human malaria, Plasmodium vivax. Brit Med J 1: 547.
- Cogswell FB (1992) The hypnozoite and relapse in primate malaria. Clin 17. Microbiol Rev 5: 26-35.
- 18. Thompson PE, Huff CG (1944) A saurian malarial parasite, Plasmodium mexicanum, N. Sp., with both elongatum and gallinaceum-types of excerythrocytic stages. J Infect Dis 74: 48–67.
- 19. Telford Jr SR (1989) Discovery of the pre-erythrocytic stages of a saurian malaria parasite, hypnozoites, and a possible mechanism for the maintenance of chronic infections throughout the life of the host. Int J Parasitol 19: 597-616.
- 20. Thayer WLotmf, p. 326. (1897) Lectures on the malarial fevers. New York: D. Appleton & Co.
- Coatney GR (1976) Relapse in malaria: an enigma. J Parasitol 62: 2–9.
 Battle KE, Karhunen MS, Bhatt S, Gething PW, Howes RE, et al. (2014) Geographical variation in Plasmodium vivax relapse. Malaria J 13: 144.
- 23. Allan RA, Mahrt JL (1989) Influence of transmission period on primary and relapse patterns of infection of Leucocytozoon spp. and Haemoproteus mansoni. Am Midl Nat 121: 341-349.
- 24. Pigliucci M (2005) Evolution of phenotypic plasticity: where are we going now? Trends Ecol Evol 20: 481-486
- 25. Scheiner SM (1993) Genetics and evolution of phenotypic plasticity. Annu Rev Ecol Syst 24: 35-68
- 26. Reece SE, Ramiro RS, Nussey DH (2009) Plastic parasites: sophisticated strategies for survival and reproduction? Evol Appl 2: 11-23.
- 27. Babayan SA, Read AF, Lawrence RA, Bain O, Allen JE (2010) Filarial parasites develop faster and reproduce earlier in response to host immune effectors that determine filarial life expectancy. PLoS Biol 8: e1000525.
- Leggett HC, Benmayor R, Hodgson DJ, Buckling A (2013) Experimental 28. Evolution of Adaptive Phenotypic Plasticity in a Parasite. Curr Biol 23: 139-142.
- 29. Reece SE, Ali E, Schneider P, Babiker HA (2010) Stress, drugs and the evolution of reproductive restraint in malaria parasites. Proc R Soc B Biol Sci 277: 3123-3129
- 30. Pollitt LC, Mideo N, Drew DR, Schneider P, Colegrave N, et al. (2011) Competition and the evolution of reproductive restraint in malaria parasites. Am Nat 177: 358–367
- 31. Hulden L, Hulden L (2011) Activation of the hypnozoite: a part of Plasmodium vivax life cycle and survival. Malaria J 10: 90.
- 32. Manwell RD (1929) Relapse in bird malaria. Am J Epidemiol 9: 308-345.
- 33. Appelgate JE (1970) Population changes in latent avian malaria infections associated with season and corticosterone treatment. J Parasitol 56: 439-443.
- Appelgate JE, Beaudoin RL (1970) Mechanisms of spring relapse in avian 34. malaria: effects of gonodropin and corticosterone. J Wildlife Dis 6: 443-447.
- 35. Pearson RD (2002) Is prolactin responsible for avian, saurian, and mammalian relapse and periodicity of fever in malarial infections? Can J Zool 80: 1313-1315.
- 36. Valkiunas G, Bairlein F, Iezhova TA, Dolnik OV (2004) Factors affecting the relapse of Haemoproteus belopolskyi infections and the parasitaemia of Trypanosoma spp. in a naturally infected European songbird, the blackcap, Sylvia atricapilla. Parasitol Res 93: 218–222.
- 37. Hulden L, Hulden L, Heliovaara K (2008) Natural relapses in vivax malaria induced by Anopheles mosquitoes. Malaria J 7: 64.
- 38. Lawaly R, Konate L, Marrama L, Dia I, Diallo D, et al. (2012) Impact of mosquito bites on asexual parasite density and gametocyte prevalence in asymptomatic chronic Plasmodium falciparum infections and correlation with IgE and IgG titers. Infect Immun 80: 2240-2246
- 39. Billingsley PF, Snook LS, Johnston VJ (2005) Malaria parasite growth is stimulated by mosquito probing. Biol Lett 1: 185-189.
- Shutler D, Reece SE, Mullie A, Billingsley PF, Read AF (2005) Rodent malaria 40. parasites Plasmodium chabaudi and P. vinckei do not increase their rates of gametocytogenesis in response to mosquito probing. Proc R Soc B Biol Sci 272: 2397-2402.

- 41. Killick-Kendrick R (1978) Taxonomy, zoology and evolution. In: Killick-Kendrick R, Peters W, editors. Rodent Malaria. London: Academic Press. pp. 1 - 52
- 42. Landau I, Chabaud A (1994) Plasmodium species infecting Thannomys rutilans: a zoological study. Adv Parasitol 33: 50-90
- 43. Roca-Feltrer A, Schellenberg JR, Smith L, Carneiro I (2009) A simple method for defining malaria seasonality. Malaria J 8: 276.
- 44. Valkiūnas G (2005) Avian Malaria Parašites and Other Haemosporidia. Boca Raton, FL., USA: CRC Press.
- 45. Frank SA (1996) Models of parasite virulence. Q Rev Biol 71: 37-78.
- 46. Alizon SA, Hurford A, Mideo N, van Baalen M (2009) Virulence evolution and the trade-off hypothesis: history, current state of affairs and the future. J Evol Biol 22: 245-259.
- 47. Beldade P, Mateus ARA, Keller RA (2011) Evolution and molecular mechanisms of adaptive developmental plasticity. Mol Ecol 20: 1347-1363.
- 48. Duneau D, Ebert ${\rm \ddot{D}}$ (2012) Host sexual dimorphism and parasite adaptation. PLoS Biol 10: e1001271.
- 49. Mideo N, Reece SE (2012) Plasticity in parasite phenotypes: evolutionary and ecological implications for disease. Future Microbiol 7: 17-24.
- 50 Van Den Berg F, Bacaër N, Metz JAJ, Lannou C, Van Den Bosch F (2011) Periodic host absence can select for higher or lower parasite transmission rates. Evol Ecol 25: 121-137.
- 51. Donnelly R, Best A, White A, Boots M (2013) Seasonality selects for more acutely virulent parasites when virulence is density dependent. Proceedings of the Royal Society B: Biological Sciences, 280(1751). Proc R Soc B Biol Sci 280: 20122464.
- 52. Koelle K, Pascual M, Yunus M (2005) Pathogen adaptation to seasonal forcing and climate change. Proc R Soc B Biol Sci 272: 971-977.
- 53. Kamo M, Sasaki A (2005) Evolution toward multi-year periodicity in epidemics. Ecol Lett 8: 378-385
- 54. Cohen D (1967) Optimizing reproduction in a randomly varying environment when a correlation may exist between the conditions at the time a choice has to be made and the subsequent outcome. J Theor Biol 16: 1-14.
- 55. Gavrilets S, Scheiner SM (1993) The genetics of phenotypic plasticity. 5. Evolution of reaction norm shape. J Evol Biol 6: 31-48.
- 56. Lande R (2009) Adaptation to an extraordinary environment by evolution of phenotypic plasticity and genetic assimilation. J Evol Biol 22: 1435-1446.
- 57. Cellier-Holzem E, Esparza-Salas R, Garnier S, Sorci G (2010) Effect of repeated exposure to Plasmodium relictum (lineage SGS1) on infection dynamics in domestic canaries. Int I Parasitol 40: 1447-1453.
- 58. Reece SE, Drew DR, Gardner A (2008) Sex ratio adjustment and kin discrimination in malaria parasites. Nature 453: 609-614.
- 59. Babiker HA, Schneider P, Reece SE (2008) Gametocytes: insights gained during a decade of molecular monitoring. Trends Parasitol 24: 525-530.
- Abdel-Wahab A, Abdel-Muhsin AMA, Ali E, Suleiman S, Ahmed S, et al. (2002) 60. Dynamics of gametocytes among Plasmodium falciparum clones in natural infections in an area of highly seasonal transmission. J Infect Dis 185: 1838-1842.
- 61. Schneider P, Schoone G, Schallig H, Verhage D, Telgt D, et al. (2004) Quantification of Plasmodium falciparum gametocytes in differential stages of development by quantitative nucleic acid sequence-based amplification. Mol Biochem Parasitol 137: 35-41.
- 62. Schneider P, Bousema JT, Gouagna LC, Otieno S, Van de Vegte-Bolmer M, et al. (2007) Submicroscopic Plasmodium falciparum gametocyte densities frequently result in mosquito infection. Am J Trop Med Hyg 76: 470-474.
- 63. Okech BA, Gouagna LC, Kabiru EW, Beier JC, Yan GY, et al. (2004) Influence of age and previous diet of Anopheles gambiae on the infectivity of natural Plasmodium falciparum gametocytes from human volunteers. J Insect Sci 4: 33.
- 64. Terzian LA, Stahler N, Irreverre F (1956) The effects of aging, and the modifications of these effects, on the immunity of mosquitoes to malarial infection. J Immunol 76: 308-313.
- Fontaine A, Diouf I, Bakkali N, Misse D, Pages F, et al. (2011) Implication of 65. haematophagous arthropod salivary proteins in host-vector interactions. Parasites Vectors 4: 187.
- 66. Titus RG, Ribeiro JM (1988) Salivary gland lysates from the sand fly Lutzomyialongipalpis enhance Leishmania infectivity. Science 239: 1306-1308.
- 67. Cameron A, Reece SE, Drew DR, Haydon DT, Yates AJ (2013) Plasticity in transmission strategies of the malaria parasite, Plasmodium chabaudi: environmental and genetic effects. Evol Appl 6: 365-376.
- 68. Carter LM, Kafsack BFC, Llinas M, Mideo N, Pollitt LC, et al. (2013) Stress and sex in malaria parasites: Why does commitment vary? Evol Med Publ Heath 2013: 135-147.
- 69. Gautret P, Coquelin F, Chabaud AG, Landau I (1997) The production of gametocytes by rodent Plasmodium species in mice during phenylhydrazine induced reticulocytosis. Acta Parasitol 42: 65-67.
- 70. Reece SE, Duncan AB, West SA, Read AF (2005) Host cell preference and variable transmission strategies in malaria parasites. Proc R Soc B Biol Sci 272: 511-517.
- 71. Paul REL, Coulson TN, Raibaud A, Brey PT (2000) Sex determination in malaria parasites. Science 287: 128-131.
- 72. Drew DR, Reece SE (2007) Development of reverse-transcription PCR techniques to analyse the density and sex ratio of gametocytes in genetically diverse Plasmodium chabaudi infections. Mol Biochem Parasitol 156: 199-209.

- Martiniere A, Bak A, Macia JL, Lautredou N, Gargani D, et al. (2013) A virus responds instantly to the presence of the vector on the host and forms transmission morphs. eLife 2: e00183.
- O'Donnell AJ, Schneider P, McWatters HG, Reece SE (2011) Fitness costs of disrupting circadian rhythms in malaria parasites. Proc R Soc B Biol Sci 278: 2429–2436.
- Hawking F, Worms MJ, Gammage K, Goddard PA (1966) The biological purpose of the blood-cycle of the malaria parasite *Plasmodium cynomolgi*. Lancet 288: 422–424.
- Mideo N, Reece SE, Smith AL, Metcalf CJE (2013) The Cinderella syndrome: why do malaria-infected cells burst at midnight? Trends Parasitol 29: 10–16.
- Greischar MA, Read AF, Bjørnstad ON (2014) Synchrony in malaria infections: How intensifying within-host competition can be adaptive. Am Nat: 183: E36–E49.
- Mayxay M, Pukrittayakamee S, Newton PN, White NJ (2004) Mixed-species malaria infections in humans. Trends Parasitol 20: 233–240.
- Valkiūnas G, Bensch S, Iezhova TA, Križanauskienė A, Hellgren O, et al. (2006) Nested cytochrome b polymerase chain reaction diagnostics underestimate mixed infections of avian blood haemosporidian parasites: microscopy is still essential. J Parasitol 92: 418–422.
- Shanks GD, White NJ (2013) The activation of vivax malaria hypnozoites by infectious diseases. Lancet Infect Dis 13: 900–906.
- Chen NH, Auliff A, Rieckmann K, Gatton M, Cheng Q (2007) Relapses of *Plasmodium vivax* infection result from clonal hypnozoites activated at predetermined intervals. J Infect Dis 195: 934–941.
- İmwong M, Snounou G, Pukrittayakamee S, Tanomsing N, Kim JR, et al. (2007) Relapses of *Plasmodium vivax* infection usually result from activation of heterologous hypnozoites. J Infect Dis 195: 927–933.
- Bensch S, Hellgren O, Perez-Tris J (2009) MalAvi: a public database of malaria parasites and related haemosporidians in avian hosts based on mitochondrial cytochrome b lineages. Mol Ecol Resour 9: 1353–1358.
- Bousema T, Griffin JT, Sauerwein RW, Smith DL, Churcher TS, et al. (2012) Hitting hotspots: spatial targeting of malaria for control and elimination. PLoS Med 9: e1001165.
- Gilden D, Mahalingam R, Nagel MA, Pugazhenthi S, Cohrs RJ (2011) Review: the neurobiology of varicella zoster virus infection. Neuropath Appl Neurobiol 37: 441–463.

- Wilson AC, Mohr I (2012) A cultured affair: HSV latency and reactivation in neurons. Trends Microbiol 20: 604–611.
- Cardona PJ, Ruiz-Manzano J (2004) On the nature of Mycobacterium tuberculosis-latent bacilli. Eur Respir J 24: 1044–1051.
- 88. Marcello A (2006) Latency: the hidden HIV-1 challenge. Retrovirology 3: 7.
- Perng GC, Jones C (2010) Towards an understanding of the herpes simplex virus type 1 latency-reactivation cycle. Interdisc Persp Infect Dis vol:2010: 262415.
- Lieberman PM (2013) Keeping it quiet: chromatin control of gammaherpesvirus latency. Nature Rev Microbiol 11: 863–875.
- Spina CA, Anderson J, Archin NM, Bosque A, Chan J, et al. (2013) An in-depth comparison of latent HIV-1 reactivation in multiple cell model systems and resting CD4+ T cells from aviremic patients. PLoS Pathog 9: e1003834.
- Kimura M, Darbro JM, Harrington LC (2010) Avian malaria parasites share congeneric mosquito vectors J Parasitol 96: 144–151.
- Lalubin F, Delédevant A, Glaizot O, Christe P (2013) Temporal changes in mosquito abundance (*Culex pipiens*), avian malaria prevalence and lineage composition. Parasites Vectors 6: 307.
- 94. Vézilier J, Nicot A, Gandon S, Rivero A (2010) Insecticide resistance and malaria transmission: infection rate and oocyst burden in *Culex pipiens* mosquitoes infected with *Plasmodium relictum*. Malaria J 9: 379.
- Griffiths R, Double MC, Orr K, Dawson RJG (1998) A DNA test to sex most birds. Mol Ecol 7: 1071–1075.
- Waldenström J, Bensch S, Hasselquist D, Ostman O (2004) A new nested polymerase chain reaction method very efficient in detecting *Plasmodium* and *Haemoproteus* infections from avian blood. J Parasitol 90: 191–194.
- Cornet S, Nicot A, Rivero A, Gandon S (2013) Malaria infection increases bird attractiveness to uninfected mosquitoes. Ecol Lett 16: 323–329.
- Huff CG (1947) Life cycle of malarial parasites. Annual Reviews in Microbiology 1: 43–60.
- Griffing SM, Kilpatrick AM, Clark L, Marra PP (2007) Mosquito landing rates on nesting American robins (*Turdus migratorius*). Vector-Borne Zoo Dis 7: 437– 443.
- 100. Crawley MJ (2007) The R Book. England: John Wiley & Sons, Ltd.